Author's response to reviews

Title: Systematic review and meta-analysis comparing land and aquatic exercise for people with hip or knee arthritis on function, mobility and other health outcomes

Authors:

Stephanie I Batterham (sibat2@student.monash.edu.au)
Sophie Heywood (physio@melbsportsmed.com.au)
Jennifer L Keating (jenny.keating@med.monash.edu.au)

Version: 8 Date: 10 May 2011

Author's response to reviews:

Formatting changes to be made:

-----------------------------

Major revisions (we require the author to make these changes)

References: References should be cited in the text using consecutive numbers in square brackets. Currently reference 28 is cited out of order

Corrected as advised

can you check the tables do not have a last line that goes over to the next page?

thanks for your help

Jenny